Author's response to reviews

Title: A pilot study of Aboriginal health promotion from an ecological perspective

Authors:

Rachel E Reilly (rachel@unimelb.edu.au)
Marion Cincotta (marionc@vaccho.com.au)
Joyce Doyle (jdoyle@bigpond.com)
Bradley R Firebrace (bradleyf@unimelb.edu.au)
Margaret Cargo (margaret.cargo@unisa.edu.au)
Gemma van den Tol (gvandentol@gvct.com.au)
Denise Morgan-Bulled (denise@vineymorganams.com.au)
Kevin G Rowley (rowleyk@unimelb.edu.au)

Version: 9 Date: 29 August 2011

Author's response to reviews:

Dear Production Team,

In the 'Authors’ contributions' section KG should have been KR. Somehow this was missed in the last re-drafting. Thank you for the opportunity to address this minor issue. We do not need to see the manuscript again prior to publication.

Many thanks,
Rachel Reilly